Author's response to reviews

Title: Quality of life after stay in surgical intensive care unit

Authors:

Fernando J Abelha (abelha@mail.telepac.pt)
Cristina C Santos (csantos@med.up.pt)
Paula C Maia (paulacmaia@portugalmail.pt)
Maria A Castro (fbee@netcabo.pt)

Version: 5 Date: 10 May 2007

Author's response to reviews:

Manuscript ID: 1760806221121740

Quality of life after stay in surgical intensive care unit
Fernando J Abelha, Cristina C Santos, Paula C Maia, Maria A Castro

To the Editorial Team:
Thank you for your work in providing the comments by Drs. Cristina Granja and Jurgen Graf. Both the reviewers helped us making a better manuscript.
The manuscript, as you suggested, was reviewed by one professional copy editing service and had been subjected to language corrections.
The following are our responses point by point to the reviewer’s reports and comments.

With best regards